Author's response to reviews

Title: Touchdown General Primer (GP5+/GP6+) PCR and optimized sample DNA concentration support the sensitive detection of human papillomavirus

Authors:

Mark F Evans (mark.evans@uvm.edu)
Christine S-C Adamson (christine.adamson@uvm.edu)
Linda Simmons-Arnold (linda.simmons-arnold@uvm.edu)
Kumarasen Cooper (kum.cooper@vtmednet.org)

Version: 5 Date: 14 November 2005

Author’s response to reviews:

The title and abstract have been checked to confirm they are identical in the submission system and manuscript.

The author list has been checked to ensure it is written in the manuscript exactly as in the submission system.

Table 2 is now one table and text references to the table have been adjusted accordingly.